**Author's response to reviews**

**Title:** Comparison of the effect of LPS and PAM3 on ventilated lungs

**Authors:**

- Hans P Hauber (hphauber@fz-borstel.de)
- Dörte Karp (dkarp@fz-borstel.de)
- Torsten Goldmann (tgoldmann@fz-borstel.de)
- Ekkehard Vollmer (evollmer@fz-borstel.de)
- Peter Zabel (pzabel@fz-borstel.de)

**Version:** 6  **Date:** 14 April 2010

**Author's response to reviews:**

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is First name Initial Last name (e.g. Joe F Bloggs)

Author list has been edited according to the guidelines.

Figures: at 100% the text of Figure 1 is not readable. We recommend you upload a higher resolution version of the figure.

Figures have been separated (fig1 and 2 in the system is fig 1A and 1B in the manuscript, fig 3-5 in the system are fig 2A-2C in the study, fig 6-8 in the system are fig 3-5 in the manuscript) and uploaded with a higher resolution.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

The manuscript has been checked for typographical errors.